Author's response to reviews

Title: MC1R variants predisposing to concomitant primary cutaneous melanoma in a monozygotic twin pair

Authors:

Cristina Pellegrini (cristina.pellegrini@cc.univaq.it)
Maria Concetta Fargnoli (mariaconcetta.fargnoli@univaq.it)
Mariano Suppa (dr.marianosuppa@gmail.com)
Ketty Peris (ketty.peris@univaq.it)

Version: 4 Date: 23 August 2012

Author's response to reviews:

L’Aquila, August 23, 2012

Dear Editor,

We are submitting the revised version of the article entitled “MC1R variants predisposing to concomitant primary cutaneous melanoma in a monozygotic twin pair” by Cristina Pellegrini, Maria Concetta Fargnoli, Mariano Suppa and Ketty Peris (MS: 1049176449747216), for consideration for publication in the section “Case reports” of BMC Medical Genetics.

As suggested by the Associate Editor, we described all gene variants reported in the text and in Figure 1 according to the standard international nomenclature of the Human Genome Variation Society and indicated changes in red in the attached revised version of the manuscript.

We appreciate your consideration of our manuscript, and we thank you in advance for your attention in this matter.

Sincerely Yours,

Ketty Peris, MD
Professor and Chairman
Department of Dermatology
University of L’Aquila
Via Vetoio, Coppito 2
67100 L’Aquila
Italy